Author’s response to reviews

Title: Postoperative Tight Glycemic Control Reduces Postoperative Infection In Patients Undergoing Surgery: A Meta-Analysis

Authors:
Yuan-yuan WANG (wayuyu@163.com)
Shuang-fei HU (hushuangfei77@sina.com)
Hui-min YING (13989356561@139.com)
Long CHEN (chenlong8420938@163.com)
Hui-li LI (lihuili1609@163.com)
Fang TIAN (fangtian212@163.com)
zhen-feng Zhou (zhenfeng9853@163.com)

Version: 5 Date: 22 May 2018

Author’s response to reviews:

Dear Ernesto Maddaloni, M.D.:

BEND-D-17-00246R2

We are so glad to hear from you. We have revised our manuscript according to the requirements. Great thanks to you and the referee for the time and effort you expend on this paper.

Best wishes.

Yours sincerely,

Dr. zhen-feng
Editor Comments:

1. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of authors Shuang-fei HU, Hui-min YING, Long CHEN, and Hui-li LI, do not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their names from the list of authors and place them in the “Acknowledgements” section instead. Please note that any changes in the list of authors requires the completion of the “Change in authorship” form.

Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group - alone - does not usually justify authorship.

We also note that Yang-fan HUO has been included in this list, however is not in the main author list. Please amend whichever is incorrect.

Answer: We are so sorry for making you confused. Any controversies of the paper were discussed and resolved by all authors. We are working as a team in dealing with this paper. So we have provided clarification on the contributions of authors as following: “Wang YY, Hu SF and Zhou ZF designed the study. Hu SF, Ying HM, Chen L and Li HL did the work of data collecting, and data was analyzed by Chen L and Tian F. Wang YY and Zhou ZF wrote the paper. Ying HM and Li HL revised the paper. Any controversies of the paper were discussed and resolved by all authors. All authors have read and approved the manuscript.” (line 18-22, page 19, Authors’ Contributions section)

Yang-fan HUO is not the main author, so we have removed.

2 Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.
Answer: We have represent authors' names using their full initials in the Authors’ Contributions section. (Refer to the first answer section)

3 Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Answer: We have included a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript. (Refer to the first answer section)

There are only two departments in this study: department of Endocrinology, Xixi Hospital of Hangzhou and department of Anesthesiology, Zhejiang Provincial People's Hospital. So if there was any revised requirement, we could easily tell all authors by face to face. Then we mailed the revised letter to every author. Finally, the paper were discussed and resolved by all authors.

4 Please ensure your figures have been uploaded in production quality. Currently Supplementary Figure 4 has not been.

Answer: We have revised Supplementary Figure 4.

5 At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: The manuscript was uploaded as a single, final, clean version that did not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files have been clean versions. Figures (and additional files) have uploaded as separate files.